Brian Staton

Semester Electricians, Patrick Tanner,

Fredonia Department of Theatre & Dance April 2016

2.29.2016 Call: Hang

6:30 PM | Tito Mercado

2:30 PM

Crew Called:

585.953.8896

Todays Call

Date:

Start:

End:

BStaton@fredonia.edu

Master Electrician Report #1

Lighting Designer: Évan Kicman Master Electrician: Brian C. Staton

Assistant Master Electrician: Patrick Tanner, Tito Mercado

Next Call

Date:	2.29.2016	Call: Hang
Start:	2:30 PM	Crew Called: Brian Staton Semester Electricians, Patrick Tanner,
End:	6:30 PM	Tito Mercado

Accomplished:

Hung, Cabled, Data

- -1st FOH
- -2nd FOH
- -1st Electric
- -2nd Electric
- -3rd Electric
- -4.4a Electric
- -5th Electric
- -Truss
- -Box Booms
- -Began Building booms

To do:

- -Finish Building Booms
- -Cable Drops for Booms
- -Check color dropped
- -Clean Cable Everywhere
- -Floor Mounts
- -Hang fixtures from Laramie
 - -1 VL2500
 - -5 LEDs
 - -4 Revolutions
- -Place Booms
- -Program pallets
- -Swich console to the eos from the element
- -Focus Show
- -Cue Show
- -Run Show
- -Strike Show

Problems Solved:

Problems---Concerns:

14 Degree was dropped from 2nd FOH. Body of instrument was damaged Barrel was fine and used in plot.

Notes: General

Notes: Focus

Upcoming Dates:

Hang 3.1.2016 | Focus 3.2.2016-3.3.2016 | Cueing 3.29.2016-3.29.2016 | Opening 4.8.2016

Fredonia Department of Theatre & Dance

April 2016

585.953.8896 BStaton@fredonia.edu

Master Electrician Report #2

Lighting Designer: Evan Kicman Master Electrician: Brian C. Staton

Assistant Master Electrician: Patrick Tanner, Tito Mercado

Todays Call

Date:		Call: Hang	
Start:	2:30 PM	Crew Called: Brian Staton Semester Electricians, Patrick Tanner,	
End:	6:30 PM	Tito Mercado	

Next Call

Date:	3.2.2016	Call: Focus
Start:	2:30 PM	Crew Called: Brian Staton Semester Electricians, Patrick Tanner,
End:	6:30 PM	Tito Mercado

Accomplished:

Finished hanging booms Added cable drops for booms Rung out System

Put floor mounts together

Cleaned Cable Hung 3rd FOH

Finished Scroller Power on 3 LX

Checked color dropped FOH and On Stage

Addressed all fixtures in space

Dropped Texture

Replaced dimmers with relays for intelligent fixtures

Spiked Trim Heights for all electrics, except 4aLX

Placed ground row

Pulled and cut color for sky cyc

Layed out focus areas

To do:

Hang Revolutions on 3LX
Hang Revolutions on 2LX
Hang VL2500 on 4aLX
Hang LEDs on 4aLX
Hang LED on 3LX
Move Booms to spike
Ring out booms
Connect booms to power

Connect booms to power Connect booms to data

Clean Cable

Lift up

Drop Color in CYC

Focus Show

Cue Show

Runs Show

Strike Show

Problems Solved:

Problems---Concerns:

Scroller that is not responding to DMX.
-Handed off to LSS and submitted tag.

Notes: General

Notes: Focus

Upcoming Dates:

Focus 3.2.2016-3.3.2016 | Cueing 3.29.2016-3.29.2016 | Opening 4.8.2016

Fredonia Department of Theatre & Dance

April 2016

Master Electrician Report #3

Lighting Designer: Evan Kicman Master Electrician: Brian C. Staton

Assistant Master Electrician: Patrick Tanner, Tito Mercado

585.953.8896 BStaton@fredonia.edu

Todays Call

Date:		Call: Focus
Start:	2:30 PM	Crew Called: Brian Staton Semester Electricians, Patrick Tanner,
End:	6:30 PM	Tito Mercado

Date:	TBD	Call: TBD	
Start:	TBD	Crew Called:	TBD
End:	TBD		

Accomplished:

Focused show Dropped color in CYC fixtures Added hazer Fixed channel 133's scroller (head and tail of scroll

was flipped) Added fixture to 3rd FOH

Repatched 3rd FOH

To do:

Next Call

Hang Revolutions on 3LX Pull up focus areas Hang Revolutions on 2LX Hang VL2500 on 4aLX Hang LEDs on 4aLX

Hang LED on 3LX

Focus LEDs on 4aLX

Focus LED on 3LX

Move Booms to spike

Ring out booms

Connect booms to power Connect booms to data

Lift up umbilicals

Patch Hazer

Focus booms

Place floor mounts

Cue Show

Runs Show

Strike Show

Problems---Concerns:

Channel 17- Check Lens

Problems Solved:

Notes: General

Notes: Focus

Upcoming Dates:

Focus 3.2.2016-3.3.2016 | Cueing 3.29.2016-3.29.2016 | Opening 4.8.2016

Brian Staton

Fredonia Department of Theatre & Dance

3.7.2016 | Call: Console Move Crew Called:

Patrick Tanner

April 2016

Todays Call

Date:

Start:

End:

585.953.8896

BStaton@fredonia.edu

2:30 PM

6:30 PM

Master Electrician Report #5

Lighting Designer: Evan Kicman Master Electrician: Brian C. Staton

Assistant Master Electrician: Patrick Tanner. Tito Mercado

Next Call

TOXI O GIII					
Date:	TBD	Call: TBD			
Start:	TBD	Crew Called:	TBD		
End:	TBD				

Accomplished:

Hang Revolutions on 3LX

Hang Revolutions on 2LX

Hang VL2500 on 4aLX

Hang LEDs on 4aLX

Hang LED on 3LX

Focus LEDs on 4aLX

Focus LED on 3LX

Move Booms to spike

Ring out booms

Connect booms to power

Connect booms to data

Ran rope light for booms

Set up shin busters

Set up Cyc scrapes

To do:

Add mirror ball specials

Lift up umbilicals

Focus booms Cue Show

Runs Show

Strike Show

Load Gel Boxes

Train Crew on Gel Boxes

Run run lights

Problems Solved:

Problems---Concerns:

Upon using the eos today. The top encoder was not responding appropriately to input. Will not effect run of show.

Notes: General/End of Show Duties

Return 10 degree frames to Eric

Notes: Focus

Focus cyc scrapes and shin busters

Hang Mirror Ball Specials

Upcoming Dates:

Cueing 3.29.2016-3.29.2016 | Opening 4.8.2016

Brian Staton

Fredonia Department of Theatre & Dance

3.7.2016 | Call: Console Move Crew Called:

Patrick Tanner

April 2016

Todays Call

Date:

Start:

End:

585.953.8896

BStaton@fredonia.edu

2:30 PM

6:30 PM

Master Electrician Report #5

Lighting Designer: Evan Kicman Master Electrician: Brian C. Staton

Assistant Master Electrician: Patrick Tanner. Tito Mercado

Next Call

TOXI O GIII					
Date:	TBD	Call: TBD			
Start:	TBD	Crew Called:	TBD		
End:	TBD				

Accomplished:

Hang Revolutions on 3LX

Hang Revolutions on 2LX

Hang VL2500 on 4aLX

Hang LEDs on 4aLX

Hang LED on 3LX

Focus LEDs on 4aLX

Focus LED on 3LX

Move Booms to spike

Ring out booms

Connect booms to power

Connect booms to data

Ran rope light for booms

Set up shin busters

Set up Cyc scrapes

To do:

Add mirror ball specials

Lift up umbilicals

Focus booms Cue Show

Runs Show

Strike Show

Load Gel Boxes

Train Crew on Gel Boxes

Run run lights

Problems Solved:

Problems---Concerns:

Upon using the eos today. The top encoder was not responding appropriately to input. Will not effect run of show.

Notes: General/End of Show Duties

Return 10 degree frames to Eric

Notes: Focus

Focus cyc scrapes and shin busters

Hang Mirror Ball Specials

Upcoming Dates:

Cueing 3.29.2016-3.29.2016 | Opening 4.8.2016

Brian Staton

Fredonia Department of Theatre & Dance April 2016

3.30.2016 Call: Cueing Day 1

Crew Called:

Patrick Tanner Tito Mercado

585.953.8896

BStaton@fredonia.edu

Master Electrician Report #6

Lighting Designer: Evan Kicman Master Electrician: Brian C. Staton

Run run lights

Assistant Master Electrician: Patrick Tanner, Tito Mercado

Next Call

Date:	3.31.2016	Call: Cueing Day 2
Start:	2:30 PM	Crew Called: Brian Staton Patrick Tanner Tito Mercado
End:	TBD	Semester Electrics Crew

Accomplished:

Todays Call

Date:

Start:

End:

Lift up umbilicals Started Cueing

- -Pas de Deux a Trois
- -Helen Meyers

2:30 PM

12:00 PM

-Pantheon Rising

Cleaned up rope light Focused Shin Busters

To do:

Add mirror ball specials Load Gel Boxes
-2 10 Degrees with Train Crew on Gel Boxes

iris's

Cue Show Runs Show Strike Show

Problems Solved:

H	rob	lems	Con	cerr	S:
٢	100	161112	COII	Cell	13

Notes: General/End of Show Duties

Return 10 degree frames to Eric

Notes: Focus

Focus cyc scrapes

Hang Mirror Ball Specials

Upcoming Dates:

Cueing 3.29.2016 | Opening 4.8.2016

Fredonia Department of Theatre & Dance April 2016

3.30.2016 Call: Cueing Day 2

Crew Called: Brian Staton

Patrick Tanner Tito Mercado

Semester Electrics Crew

585.953.8896

Todays Call

Date:

Start:

End:

BStaton@fredonia.edu

2:30 PM

12:00 PM

Master Electrician Report #7

Lighting Designer: Evan Kicman Master Electrician: Brian C. Staton

Assistant Master Electrician: Patrick Tanner, Tito Mercado

Next Call

Date:	3.31.2016	Call: Run Cre	w Training
Start:	6:00 PM	Crew Called:	Brian Staton
End:	TBD		

Accomplished:

Added mirror ball specials -2 10 Degrees with iris's **Cued Show Load Gel Boxes**

To do:

Prep Paper Frames Train Crew on Gel Boxes Run run lights **Runs Show** Strike Show

Problems Solved:

VL Lamp Blew. Lamp was changed by LLS

Problems---Concerns:

Notes: General/End of Show Duties

Return 10 degree frames to Eric

Notes: Focus

Focus cyc scrapes (if needed)

Upcoming Dates:

All Day Tech 4.1.2016 Opening 4.8.2016

Fredonia Department of Theatre & Dance

Master Electrician Report #8

Lighting Designer: Evan Kicman Master Electrician: Brian C. Staton

Assistant Master Electrician: Patrick Tanner, Tito Mercado

April 2016 585.953.8896

BStaton@fredonia.edu

Todays Call

Date:		Call: Run Cre		
Start:	2:30 PM	Crew Called:	Brian Staton	
End:	12:00 PM			

Next Call

Date:	3.31.2016	Call: ALL DAY TECH
Start:	6:00 PM	Crew Called: Brian Staton Patrick Tanner Tito Mercado
End:	TBD	rather famile into mercado

Accomplished:

Train Crew on Gel Boxes Ran run lights Ran video monitors -Dana -Backstage

Ran Stage Manager Cue list monitor

To do:

Put noxboom Chairs Back Take up rope for boxbooms Prep Paper Frames Train Crew on Gel Boxes Run run lights Runs Show Strike Show

	_				
Prol	\cap	2ms	20	VQC	٠

Problems---Concerns:

Notes: General/End of Show Duties

Return 10 degree frames to Eric

Notes: Focus

Focus cyc scrapes (if needed)

Upcoming Dates:

All Day Tech 4.1.2016 Opening 4.8.2016

Fredonia Department of Theatre & Dance April 2016

585.953.8896

BStaton@fredonia.edu

Master Electrician Report #9

Lighting Designer: Évan Kicman Master Electrician: Brian C. Staton

Assistant Master Electrician: Patrick Tanner, Tito Mercado

Todays Call Next Call

Date: 4.3.2016 Call: Dress Rehearsal

Start: 4:15 PM Crew Called: Brian Staton
Patrick Tanner Tito Mercado

Start: 5:00 PM End: TBD

4.4;.2016

Call: Dress Rehearsal 2
Crew Called: Brian Staton
Patrick Tanner Tito Mercado

Accomplished:

End:

Prepped Paper Frames Changed lamp in channel 84

9:00 PM

To do:

Strike Show

Date:

Add diffusion to Box Booms Trouble shoot channel 108 (see concerns) Clean House Runs Show

Problems Solved:

Problems---Concerns:

Channel 108, Colorado Zoom Tour 2, does not zoom

Notes: General/End of Show Duties

Return 10 degree frames to Eric

Notes: Focus

Focus cyc scrapes (if needed)

Upcoming Dates:

Fredonia Department of Theatre & Dance

Call: Dress Rehearsal 2

Crew Called: Brian Staton

Patrick Tanner Tito Mercado

April 2016

Todays Call

Date:

Start:

End:

585.953.8896

BStaton@fredonia.edu

4.4.2016

5:00 PM

10:00 PM

Master Electrician Report #10

Lighting Designer: Evan Kicman Master Electrician: Brian C. Staton

Assistant Master Electrician: Patrick Tanner, Tito Mercado

Next Call

Date:		Call: Dress Rehearsal 3
Start:	2:30 PM	Crew Called: Brian Staton Patrick Tanner Tito Mercado
Fnd [.]	TRD	ratrick fairner filo Mercado

Accomplished:

Changed front light color

-R02 to NC

-R65

Added Diffusion to box booms

Added monitor for Greenroom

Added piece number lables on gel boxes

Replaced house lights (Done by LSS)

Linked all cue lists

To do:

Trouble shoot channels 108 and 112 (see concerns)

Move Console

Add tophats to VL2500s

Check all parameters of fixtures during channel check

Tape the sides of paper frames for booms

(prevent gel from falling out during changes)

Clean House

Runs Show

Strike Show

Problems Solved:

Problems---Concerns:

Channels 108 and 112, Colorado Zoom Tour 2, do not zoom

Notes: General/End of Show Duties

Return 10 degree frames to Eric

Notes: Focus

Refocus sky cycs

Refocus ground row

Check focus on upstage right boom (was bumped during warm ups)

Upcoming Dates:

Fredonia Department of Theatre & Dance

April 2016

Todays Call

Date:

Start:

End:

585.953.8896

BStaton@fredonia.edu

4.5.2016

5:00 PM

10:00 PM

Master Electrician Report #11

Lighting Designer: Evan Kicman Master Electrician: Brian C. Staton

Assistant Master Electrician: Patrick Tanner, Tito Mercado

Next Call

Date:		Call: Dress Rehearsal 4
Start:	2:30 PM	Crew Called: Brian Staton Patrick Tanner Tito Mercado
End:	TBD	Tatrick farmer Tito Mercado

Accomplished:

Focused the Sky Cyc cells down
Added full top hats to the VL2500s
Fixed focus on the ground cyc
Re-Focused 3rd front of house offstage fixtures
Taped the sides of paper frames to avoid from gels
falling out

Call: Dress Rehearsal 3

Crew Called: Brian Staton

Patrick Tanner Tito Mercado

Tagged 108 and 112 (see concerns) Replaced lamp in ghost light Replaced bace on channel 82

To do:

Fix 'select' button
Add box fan to hazer
Add color and diffusion to box boom fixtures for a
curtain warmer

Move Console Clean House Runs Show Strike Show

Problems Solved:

LSS: Channel 82 had bad bace tagged on LSS Desk,EH12.

Problems---Concerns:

Channels 108 and 112 seem to function properly tagged to not go to peach after.

Notes: General/End of Show Duties

Return 10 degree frames to Eric

Notes: Focus

Upcoming Dates:

Fredonia Department of Theatre & Dance April 2016

585.953.8896

BStaton@fredonia.edu Todays Call

4.6.2016

5:00 PM

10:00 PM

Master Electrician Report #12

Lighting Designer: Evan Kicman Master Electrician: Brian C. Staton

Assistant Master Electrician: Patrick Tanner, Tito Mercado

Next Call

Date:	4.7.2016	Call: Dress Rehearsal 5	
Start:	2:30 PM	Crew Called: Brian Staton Patrick Tanner Tito Mercado	
End:	TBD	ratiick fairifer Tito Wercado	

Accomplished:

Date:

Start:

End:

Added box fan to hazer

Add color and diffusion to box boom fixtures for a

curtain warmer

Call: Dress Rehearsal 4

Crew Called: Brian Staton

Patrick Tanner Tito Mercado

To do:

Runs Show Strike Show

Replace burnt out boom gel in booms
Add black wrap to scrollers on the 4 and 4a electric
Swap channel 108 and 109
Fix 'select' button
Move Console
Clean House

Problems Solved:

Problems---Concerns:

Channels 108 and 112 did not zoom during run of show today

Notes: General/End of Show Duties

Return 10 degree frames to Eric

Notes: Focus

Upcoming Dates:

Brian Staton

Fredonia Department of Theatre & Dance

April 2016

585.953.8896 BStaton@fredonia.edu

4.7.2016

5:00 PM

10:00 PM

Master Electrician Report #13

Lighting Designer: Evan Kicman Master Electrician: Brian C. Staton

Assistant Master Electrician: Patrick Tanner, Tito Mercado

Next Call

Date:	4.8.2016	Call: Work Notes/Opening Night
Start:	2:30 PM	Crew Called: Brian Staton Patrick Tanner Tito Mercado
End:	TBD	rattick faither tho Mercado

Accomplished:

Todays Call

Date:

Start:

End:

Added black wrap to scrollers on the 4 and 4a electric Swapped channel 108 and 109 Move Console Put the Fans on a constant

Call: Preview
Crew Called:

Patrick Tanner Tito Mercado

To do:

Tape downstage side of boom rope light Replace burnt out boom gel in booms Fix 'select' button Clean House Runs Show Strike Show

	1		
Prob	10mc		1100.
	10.111.5	,)()	1 / / / . ()

Problems---Concerns:

Notes: General/End of Show Duties

Return 10 degree frames to Eric

Notes: Focus

Upcoming Dates:

Fredonia Department of Theatre & Dance April 2016

585.953.8896 BStaton@fredonia.edu Master Electrician Report #14

Lighting Designer: Evan Kicman Master Electrician: Brian C. Staton

Assistant Master Electrician: Patrick Tanner, Tito Mercado

Todays Call

Date:		Call: Work Notes/Opening Night
Start:	5: 00 PM	Crew Called: Brian Staton Patrick Tanner Tito Mercado
End:	10:00 PM	Tatrick farmer into Welcado

Next Call

Date:	4.8.2016	Call: Strike
Start:	2:30 PM	Crew Called: Brian Staton Patrick Tanner Tito Mercado
End:	TBD	Semester Electrics Crew

Accomplished:

Tape downstage side of boom rope light Replace burnt out boom gel in booms Touched up focus on LEDs Touched up focus on Texture Back

To do:

Fix 'select' button Clean House Runs Show Strike Show

Problems Solved:

Problems---Concerns:

Power connection from channel 103 was loose connection reseated. Power to 102 and 101 was then functioning

Notes: General/End of Show Duties

Return 10 degree frames to Eric

Notes: Focus

Touched up focus on LEDs

Touched up focus on Texture Back

Upcoming Dates:

Strike 4.11.2016